DCS and TDAQ for AFP

<u>Elżbieta Banaś</u>, Krzysztof Korcyl on behalf of the ATLAS Forward Proton DCS and TDAQ Groups



The Henryk Niewodniczański Institute of Nuclear Physics Polish Academy of Sciences Cracow, Poland

LHC Forward Physics Working Group

CERN, 15-16 March 2016

Outline

o DCS

- status after installation in the tunnel
- plans
- o L1 Trigger
 - full system project
 - status after installation in the tunnel
 - plans
- o DAQ
 - full system project
 - status after installation in the tunnel, status in SR1
 - plans

DCS: overview of installed AFP Hardware Structure



15-16/03/2016

Low Voltage Supply

1st stage: USA 15

- Wiener PL512, Ethernet controlled , OPC DA server in VM2008
 Framework software + technical panels **READY**
- LVPP 4 (IBL like) split each voltage line to 4 lines, with precise current measurement ELMB, OPC UA server
 Framework software + technical panels READY (2)

2nd stage: tunnel

Voltage Regulators Crate VREG (IBL like)

- 2 LV boards for supply FE-I4 chips in Near and Far Stations
- \circ ~ 1 V_{VDC} board for optoboard
- \circ 1 controller board
- Radiation hard voltage regulators,

control – ELMB with non standard firmware, OPC UA server

IBL software – technical panels **READY** (...

15-16/03/2016

DCS and TDAQ for AFP





High Voltage Supply, Optoboard power supply

ISEG HV System:

- Crate: ECH 238 1200W-UPS \bigcirc
- FHS F405n 16 channels module \cap CAN-controlled. OPC UA server (new design)

Framework software + technical panels **READY** (...)



02/12/2016 11:00:01 PM 02

03/13/2016 11:03:02 PM.642 ATLAFPLCS:ELMB/USA15_BUS/SCC 2.41

SCOL – power supply for optoboard (IBL-like)

- 3 different supply voltages:
 - \circ **V**_{VDC} main supply (1st stage)
 - V_{pin} depletion voltage for the PIN receiver diode
 - **V**_{iset} operation point of the VCSELs
- **Optoboard Reset Signal**

Framework software + technical panels **READY**



lseg Mo	dule Op	bera	tion										
Module Nam	e: ATLAF	PLCS	:lseg/o	can0	/crate02	/ma	00	N	ode ID:		0		
Model	All (16	Chan	nels)	Dev	/ice ID		0	D	evice Cl	ass	26		
- Channels -		ch	VMea	IS	IMeas		Vset		Iset				
	ON/OFF	00	0.00	V	-0.000	A	1.00	٧	0.0001	А	Set		
near	ON/OFF	01	0.01	۷	-0.000	A	1.00	v	0.0001	A	Set		
station	ON/OFF	02	0.00	V	-0.000	A	1.00	٧	0.0001	А	Set		
	ON/OFF	03	0.00	V	-0.000	A	5.00	v	0.0001	A	Set		
	ON/OFF	04	0.00	V	-0.000	A	1.00	V	0.0001	A	Set		
far	ON/OFF	05	0.01	V	-0.000	A	1.00	۷	0.0001	А	Set		
station	ON/OFF	06	0.01	V	-0.000	A	1.00	V	0.0001	А	Set		
	ON/OFF	07	-0.00	V	-0.000	A	1.00	V	0.0001	A	Set		
- Commands Switch Cha	s for all cha innels ON a	annels / OFF	On		Off	F	General RampSp	l Inf	ormatior I	1.00) %V	n	
			_						-		_		- Optoboar
	are a	10140		1101-			(A) - A1	_ `					VPin
	Time F	armu . Range :	Y Axe	vPin s ▼	Save	Othe	er 🔻 1:1	ie) T k	og 🦵 auto				VM0
	-							-					Analog
			••	•	•••	•••	•••	•	•••	•••	••		Po
													V
		•									•	•••	VSet



Interlock Matrix Crate and temperature monitoring

All power supplies are interlockable



Temperature sensors - 45 for one Arm:

detector, heat exchanger, pot wall, stepper motor, Air Cooler, Local Trigger Board, optoboard, VREG -> connected to Interlock Matrix Crate

••

- Crucial sensors Hardware interlock action
 Defined and tested **READY**
- Monitoring

ELMB, CanOpen UA OPC server

Framework software + technical panels **READY** (

	IMC - temperat	ure meas	urement			AFP Arn	n C
	ELMB2/PP2B/C	Name:	PTOB 4	Serial Numb	oer: State: Operational 5 More details		
ſ	- Values		Alarm settin	gs: se	everities, ra	anges and a	acknowledgement
	FS NTC P0	31.17 C	ACTIVE	Α	in range	W 34.00	E AFP ARMC FAR SIT1 TEMP PDModule: (NoName)
	FS NTC P1	31.67 C	ACTIVE	Α	in range	W 34.00	E Time Range * Y Axes * Save Other * 1:1 bg auto
	FS NTC P2	30.99 C	ACTIVE	Α	in range	W 34.00	E 9-
	FS NTC P3	31.14 C	ACTIVE	Α	in range	W 34.00	
	NS NTC P0	28.45 C	ACTIVE	Α	in range	W 35.00	
	NS NTC P1	28.65 C	ACTIVE	Α	in range	W 35.00	
	NS NTC P2	27.22 C	ACTIVE	Α	in range	W 34.00	E Thu 03 Mar 2016 07:00:00 AM CET Thu 03 Mar 2016 02:00:00 PM CET
	NS NTC P3	-97.60 C	ot configured	1			03/13/2016 11:07:02 PM 922 ATLAFPLCS:ELMB/USA16_BUSIMC, 20:18
	OPTOBOARD NTC C	22.17 C	ot configured	1			

15-16/03/2016

Secondary Vacuum and Cooling



Movement and Positioning control



Figure 2. The IN, OUT, and HOME switches and their ideal ranges settings. In addition the electrical STOP needle switch (and photo insert) is shown and its relation to the positions of the other switches.

2

Covered by Roman Pot Instrumentation and Control system - Xavier Pons (PH-DT)

Last hardware commissioning tests done

Integration with DCS – to be started soon

08MAR2016

AFP-TM

Courtesy of M. Rijssenbeek, AFP Technical Meeting, 08.03.16

15-16/03/2016

Goal:

internal FE chips parameters (temperature, supply voltage, current, ...) are collected by DAQ and transferred to DCS



OPC UA server embedded in RCE will acquire health parameters via dedicated API's:

- OPC UA server framework installed on RCE, ready for development
- \circ $\,$ Access to test station for API development is setup
- \circ $\,$ Calibration constants will be provided by detector experts $\,$

Status - ongoing 💙

DCS Status and plans

•••

- DCS production project is setup in Point1 environment \bigcirc
 - Archiving \bigcirc
- Communication protocols, hardware access **READY** \bigcirc
 - OPC UA servers in SLC6: HV and CanOpen \bigcirc
 - OPC DA server in VM 2008: LV Wiener PL512
 - S7 Native Driver for PLC S7 1200 control
- Hardware Interlock implemented READY 😬
- Technical panels for detector powering **READY** \bigcirc
- Technical panels for vacuum **READY** 🙂 for Cooling ongoing 💔 \bigcirc
- Integration with Movement an Positioning System to be done 🚫 \bigcirc
- FSM started, ongoing Ο
- Automatic emergency actions- software interlock started, ongoing \bigcirc
- Notification system started, ongoing 🏹 Ο
- Integration with ATLAS Central DCS to be done 🚫 Ο



Documentation.... 15-16/03/2016

AFP LVL1 trigger



AFP2+0 LVL1 trigger





AFP LVL1 trigger status and development

• AFP LTB boards commissioned and installed in the tunnel

• trigger signals from the LTBs arrive at CTP in decent form (after 250 m of air-core cables)

- NIM logic in CTP crate discriminates signals to make 23 ns pulses
- copies for standalone runs arrive at AFP rack and can be used as triggers
 - as single stations or in coincidence
- AFP LVL1 trigger ready for commissioning

AFP DAQ





Status of standalone AFP TDAQ

ATLAS TDAQ SOFTWARE - Partition AFP_SR1	@ CD1
File Commands Access Control Settings Logging Level Help	
Commit & Reload Reload Panels +	Controls LTP+TTC in VME crate
RUN CONTROL STATE RUNNING Run Control Segments & I	Resources Datase
- Run Control Commands	Controls RCE to configure and run EE-14 🚫
SHUTDOWN INITIALIZE Setup enabled	
UNCONFIG CONFIG CONFIG	C-SR1 enabled Monitoring data quality via sampler @ DCM 🚫
STOP START	nabled
HOLD TRG RESUME TRG	sy enabled
Beam Stable 🕘 Ready for Physics 🏐 🧐 🖓 🛶 🖓 AFP-BusyChar	
Run Information & Settings	Firmware at HSIO controlled from RCE
Lumi Block 0	• interfaces with TTC
Number Rate	enabled • L1A, BCR, ECR
Level 1 0 0.00 mHz	• requests FE data, formats fragments then forwards to ROS
HLT 0 0	 copies of events sent to RCE for monitoring
Recorded 0 0	• test data files written via ROS with full ATLAS formatting
Information Counters Settings	

• fast trigger cables reach CTP rack; signals discriminated, copies sent to AFP rack

• CTP signal cable between CTP and AFP racks arrived, but needed to be tested

OVER SIGNAL CONTROL Set Ween of a narried set ween of a narried set needed to be tested • TTC system from SR1 moved to USA15 for commissioning detectors in the tunnel

• OKS configuration for standalone partition ready for data from detectors - waiting for ROS PC

standalone programs calibGui and cosmicGui will be used in commissioning until partition starts controlling RCE
 15-16/03/2016 DCS and TDAQ for AFP 17